



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL079807

Key Points:

- Precipitation variability is projected to increase globally
- Northern hemisphere midlatitudes are a key region for worsening pluvials
- Drying regions may see more pluvials that are longer and stronger

Supporting Information:

Supporting Information S1

Correspondence to: E. R. Martin, elinor.martin@ou.edu

Citation:

Martin, E. R. (2018). Future projections of global pluvial and drought event characteristics. *Geophysical Research Letters*, 45, 11,913–11,920. https://doi.org/10.1029/2018GL079807

Received 27 JUL 2018 Accepted 13 OCT 2018 Accepted article online 19 OCT 2018 Published online 4 NOV 2018

Future Projections of Global Pluvial and Drought Event Characteristics

E. R. Martin^{1,2}

¹School of Meteorology, University of Oklahoma, Norman, OK, USA, ²South Central Climate Adaptation Science Center, Norman, OK, USA

Abstract This study assesses projections from 24 CMIP5 models of number, duration, and severity of pluvial and drought events utilizing 6-month standardized precipitation index. Increased variability of standardized precipitation index is projected globally. More frequent, longer lasting, and stronger pluvials are projected in wet regions, and the same for droughts in dry regions. Worsening pluvials and droughts are most apparent in the Northern Hemisphere midlatitudes and the Americas, respectively. Uniquely, this study investigates pluvials and droughts in locations where the precipitation trend is of the opposite sign. In drying regions, 40% of grid points project an increase in number and 65% project an increase in duration of severe pluvials. Projections for severe drought events in wetting regions show similar projections. As precipitation trends alone do not provide information about pluvial and drought characteristics this study has important implications for planning and resilience.

Plain Language Summary The variability of precipitation is projected to increase globally in the future, which has a multitude of impacts including on water resources, agriculture, public health, and fire outbreaks. This study uses future projections from global climate models to evaluate how the number, duration, and severity of extended wet periods (pluvials) and dry periods (drought) will change in the future for the first time. More frequent, longer lasting, and stronger pluvials are projected in wet regions of the world, and the same for droughts in dry regions. The Americas, including Central America, the Caribbean, and the Amazon, are a hot spot for worsening droughts, and northern North America and Europe are hot spots for worsening pluvials. Uniquely, this study investigates pluvials and droughts in locations where the precipitation trend is of the opposite sign. For the most severe pluvial events, almost half of locations with a drying trend showed an increase in number of pluvials and 65% of locations showed an increase in duration. Projections for severe drought events in wetting regions show similar projections. As mean precipitation trends alone do not provide information about pluvial and drought characteristics this study has important implications for planning and resilience.

1. Introduction

Pluvial and drought events have large socioeconomic impacts around the world. Impacts from excessive rainfall, or pluvials, include human and agricultural loses from flooding, waterborne disease outbreaks, and the buildup of vegetation that can be a fire hazard in later seasons (e.g., Auld et al., 2004; Bronstert, 2003; Govender et al., 2006). Impacts from drought include reduced surface and ground water resources, crop failure, and tree mortality (e.g., Anderegg et al., 2013; Epule et al., 2014; Li et al., 2009). With increased precipitation variability in a warmer climate (e.g., Pendergrass et al., 2017; Räisänen, 2002; Rind et al., 1989), changes in characteristics of pluvial and drought events (e.g., number, duration, and severity) might be expected. In order to increase resilience to these events, it is necessary to understand how these wet and dry periods will change in the future.

Much prior work has focused on changes in drought characteristics. Recent studies, including those using models from the Coupled Model Intercomparison Project Phase 5 (CMIP5; Taylor et al., 2012, show increasing drought frequency, especially for the most extreme droughts in subtropical regions (e.g., Cook et al., 2014; Dai & Zhao, 2017; Hunt, 2011; Orlowsky & Seneviratne, 2013; Touma et al., 2015). However, there has been much less focus on how the characteristics of extended wet periods, or pluvials will change in the future. While we expect the most intense rain events to increase (e.g., O'Gorman & Schneider, 2009), it is not known if this will

©2018. American Geophysical Union. All Rights Reserved. be true for pluvials, like the May 2015 heavy rainfall event in Oklahoma and Texas. May 2015 was the wettest month on record (since 1895) for both states and was a the major cause for Oklahoma and Texas to become drought free for the first time since 2010 (NOAA, 2016). Studies that have looked at droughts and pluvials together such as Kangas and Brown (2007), highlighted that the largest annual drought and pluvial events occur more frequently in the central United States, and drought-pluvial dipoles have also been highlighted in the Great Plains (Christian et al., 2015). Duffy et al. (2015) investigated dry and wet periods in the Amazon and noted that the characteristics of projected future drought and pluvials are best understood in the context of projected changes in mean precipitation.

As there have been no global studies of how the characteristics of pluvial events at less than interannual scales are projected to change in the future, this study quantifies those changes with the CMIP5 archive and makes comparisons with changes to drought characteristics. Importantly, this study will put changes in frequency, severity, and duration of pluvials in the context of future precipitation trends. A description of the data sets and definitions are found in section 2, with results for the characteristics of pluvial and drought events shown in section 3.

2. Data and Methods

Observations of precipitation from two land-only monthly data sets are used for the historical period; Climate Research Unit (CRU) TS v3.23 precipitation at 0.5° resolution from 1901 to 2014 (Harris et al., 2014) and Global Precipitation Climatology Center (GPCC) v7 precipitation at 1° resolution from 1901 to 2015 (Schneider et al., 2011). Historical and future projection (representative concentration pathway; RCP4.5 and RCP8.5) simulations are used from 24 CMIP5 models (see Table S1 in the supporting information for model details), with one ensemble member per model used. Results from the more moderate RCP4.5 scenario are similar, but magnitudes are smaller (see supporting information). Model horizontal resolution ranges from 0.75° to 2.8°, and data are regridded bilinearly to a common 2° horizontal resolution between 70 N and S to facilitate comparison, as is commonly done in multimodel analysis (e.g., Martin et al., 2014). Observational data sets are used at their native resolution. Changes in future characteristics are established using differences between 2080-2100 RCP output and 1980–2000 historical output. Statistical significance is measured using a Student's *t* test at the 95% level for observations, and model spread assessed using the number of models that agree on the sign of the trend.

There is no singular definition of pluvials nor droughts (Wilhite & Glantz, 1985). To identify meteorological pluvial and drought events, this study uses the 6-month standardized precipitation index (SPI; McKee et al., 1993). SPI is calculated at each grid point, for each observational data set and model, for the duration of the observations (1901–2015) and the model output (1900–2100). SPI has thresholds for precipitation deficits and excesses, making it of particular use for this study. Pluvial (positive) and drought (negative) and events are defined based on exceedence of two thresholds: moderate (\pm 1) and severe (\pm 2). For each category (moderate or severe) and for pluvials and droughts, we define (1) an event occurrence as a continuous period of time when the SPI exceeds that level, (2) the duration as the length of an event in months, and (3) the intensity as the average SPI during an event. As the 6-month SPI is used, the thresholds only need to be exceeded for 1-month to be identified as an event. As some water management decisions are influenced by the intensity and duration of an event, we calculate a "severity index" (SI), based on Peters (2014), which combines the two variables. The index is given by

$$SI = \frac{\sum_{i=1}^{n} SPI_i \times T_{event}}{\overline{SPI} \times \overline{T}}$$
(1)

where SPI_i is the monthly SPI during the event, T_{event} is the number of months during which the SPI exceeds the appropriate threshold, \overline{SPI} is the mean value of all the months that SPI exceeds the threshold, and \overline{T} is the mean duration of all the events that exceed the threshold.

3. Results

3.1. Climatology and Trends

In the 20th century, both observational data sets show similar trends in precipitation across the globe (Figure S1). Significant increases in precipitation are observed across the midlatitudes and decreasing trends across large sections of the subtropics. The CMIP5 models represent the general spatial patterns of the observed trends in the historical period. Projections for the 21st century (Figure 1a), continue the observed



Figure 1. Multimodel mean trends (RCP8.5 2006–2100) in (a) precipitation and (b) 9-year running standard deviation in SPI. Stippling indicates regions where at least 18 of the CMIP5 models agree on the sign of the trend. SPI = standardized precipitation index; RCP = representative concentration pathway; CMIP5 = Coupled Model Intercomparison Project Phase 5.

trends of wetting in the deep tropics and midlatitudes, and drying in the subtropics, with a large number of models agreeing on this pattern as seen in prior studies (Collins et al., 2013).

The distribution of SPI from CRU, GPCC, and CMIP5 historical simulations is shown in Figure 2 and illustrates the historical simulations capability in replicating the observed probability distribution function (results are similar for land only as shown in Figure S2). The GPCC and CMIP5 historical multimodel mean have narrower distributions than CRU, which is likely due to the higher spatial resolution of CRU being able to resolve more extremes. In the future, the SPI mean shifts to the right (becoming wetter) and widens (more extremes), implying more variability.

Despite the model and observations agreeing on historical trends in mean precipitation (and SPI, not shown), the trend in variability of SPI shows a very different spatial signal. As shown in Figure 1b, interannual precipitation variability, as measured by the 9-year running standard deviation of the SPI, shows an increasing



Figure 2. Probability distribution functions of standardized precipitation index from observations (brown and blue) and CMIP5 simulations (black) of the past (solid) and future (dashed) for all grid points. Observational and historical CMIP5 simulations are for 1980–2000 and future CMIP5 simulations (RCP8.5) are 2080–2100. CMIP5 = Coupled Model Intercomparison Project Phase 5; RCP = representative concentration pathway; CRU = Climate Research Unit; PDF = probability density function; GPCC = Global Precipitation Climatology Center.

trend across almost the entire globe in the 21st century, even in regions that are projected to become drier, such as Central America, the Intra-America Seas, Southern Africa, and the southwest United States (Figure 1a). The largest increases in variability, with highest model agreement, are seen in the tropics and midlatitudes, and the smallest increases (and decreases in some locations) with less model agreement, are in the subtropical ocean basins where precipitation is low. This pattern matches well with those shown in Pendergrass et al. (2017), which uses a different definition of variability. The increasing variability globally suggests more frequent pluvial and drought events will occur in the future.

Interestingly, there is much less agreement in the trend of SPI variability between the two observational data sets, and the CMIP5 historical simulations (Figure S2). The CMIP5 historical multimodel mean shows almost zero variability trend in the 20th century, whereas the observations show large trends, but they do not agree with each other in many locations. Across the US, both observations show increased variability (except in the south central United States), but in some locations, northern Canada and central Asia in particular, the trends are large and opposite. Agreement is better when observational data sets are limited to 1930 onwards (not shown) but disagreements in sign are still evident. Differences in variability at multiple timescales data sets was also highlighted in a review paper by (Sun et al., 2018) and attributed to sparsity of observations and interpolation techniques. This highlights the challenges in understanding precipitation variability at interannual and longer timescales.

In the historical period, the observational data sets agree on the spatial patterns of pluvial and drought characteristics (Figures S3 and S4). Duration and average SPI of pluvial and drought events are more spatially





-100 -80 -60 -40 -20 0 20 40 60 80 100 %

b) Length of Drought Events



e) Number of Pluvial Events



-100 -80 -60 -40 -20 0 20 40 60 80 %





-100 -80 -60 -40 -20 0 20 40 60 80 100



g) Avg. SPI of Pluvial Events

%

80 100

-100 -80 -60 -40 -20 0 20 40 60 80



-50 -40 -30 -20 -10 0 10 20 30 40 50



% 50 -40 -30 -20 -10 0 10 20 30 40 50

-50 -40 -30 -20 -10 0 10 20 30 40 50 Figure 3. Coupled Model Intercomparison Project Phase 5 multimodel mean percentage changes in drought (left) and

pluvial (right) moderate event characteristics: (a and e) number, (b and f) length, (c and g) intensity, and (d and h) severity index. Changes are calculated as future (RCP8.5 2080-2100) to historical (1980-2000). Note the reversed color scales for drought and pluvials characteristics. RCP = representative concentration pathway.

uniform than the number of events, which reflects in the severity index. Regions with the largest variability, as indicated by a high number of pluvial and drought events, extend across the subtropics including southern and eastern Brazil, the Sahel, and India. On average, the longest lasting pluvial and drought occur in northern South America and the horn of Africa (over 4 months for moderate events and 2 to 3 months of severe events). Apart from the deserts of the Sahara, central Asia and the middle East, the average SPI of events is relatively constant across land regions in observations.

There are distinct similarities and differences in the CMIP5 multimodel mean representation of historical pluvial and drought characteristics. The number of pluvial events are underestimated in the models in midlatitudes and drought events are overestimated (by 100% in some locations such as northern North America). The duration and average intensity of events are captured more realistically in the CMIP5 models, but models do miss the enhancement of duration in locations such as northern South America. The regions with largest disagreements between the models and observations are also those with largest disagreements





Figure 4. Multimodel mean changes between (RCP8.5 2080–2100) and historical (1980–2000) number and duration of drought/pluvial events at four different thresholds for all grid points (a–d). Each marker indicates an individual grid point. Grid points are isolated for regions with multimodel mean increasing precipitation trends (middle (e)–(h), blue) and decreasing trends (bottom (i)–(l), brown). Numbers indicate fraction of grid points in each quadrant. RCP = representative concentration pathway.

between the two observational data sets in the variability trends (Figure S2). Although the similarities in the spatial distribution of pluvial and drought event characteristics in the CMIP5 models provide confidence in future projections, biases in the magnitudes (especially for number of events) must be considered when interpreting the future projections

3.2. Changes in 21st Century Drought and Pluvial Characteristics

The multimodel mean percentage changes in pluvial and drought event characteristics are shown in Figure 3 for moderate events (Figure S5 for severe events). For moderate events, the number of pluvials are projected to increase by well over 100% in the midlatitudes including across the northern half of North America and most of Asia. The number of drought events is projected to increase by at least 50% across the subtropics, with the largest increases over land in Brazil, surrounding the Mediterranean, and southern Africa. Generally, increases in number of pluvial events are colocated with wetting regions and increases in number of drought events are colocated with wetting regions and increases in number of drought follow the same spatial pattern as the changes in number of events. Increased pluvial duration locations include Alaska and interior Asia and drought duration locations include the Caribbean and Mediterranean (increases of approximately 100% or 2–3 months).

Geophysical Research Letters



Figure 5. Regions showing opposite precipitation trends and drought/pluvial characteristics occur for (a) number, (b) duration, (c) average SPI, and (d) severity index of severe events for RCP8.5. Stippling indicates regions where at least 18 models agree on the sign of the precipitation trend. SPI = standardized precipitation index; RCP = representative concentration pathway.

The intensity of pluvials and droughts is projected to increase almost uniformly across the globe, irrespective of whether the precipitation is increasing or decreasing, with intensity increasing approximately 10%. Hence, both wet and dry events will likely become more intense in the future, even in regions where the number and/or duration of the events are decreasing. Changing duration and intensity are both captured in the severity index for moderate events, with the maximum increase in pluvial event severity approximately 67%, compared to a maximum increase of 43% for drought severity. Although the pluvial and drought changes almost mirror each other, as expected through of the "rich-get-richer" or "wet-get-wetter" mechanism (e.g., Chou et al., 2009; Trenberth, 2008), there are regions that are projected to experience increases in pluvial and drought characteristics (Table S2). This is most apparent for severe events, where 43% of grid points are projected to see increased duration of pluvials and droughts.

To compare projected changes in event characteristics, Figures 4a-4d show changes in number versus duration for each grid point in the CMIP5 multimodel mean. For pluvial thresholds, the most common quadrant is the upper right, indicating increased number and duration of pluvial events. For severe pluvials, 72% of grid points fall in this quadrant. For the most severe droughts, almost 42% of grid points fall in this quadrant. However, for the moderate drought threshold, the most common quadrant (almost 51% of grid points) is a reduced number and duration of droughts. It is likely that this is because with increased intensity of droughts, they are being "upgraded" from the moderate to severe category, although it is interesting to note that this does not happen on the pluvial side.

To establish whether pluvials and droughts are occurring in conjunction with increasing and decreasing precipitation trends respectively (i.e., rich-get-richer), Figures 4e–4h present the locations only where the multimodel mean precipitation trend is increasing and Figures 4i–4l present the locations only where the multimodel mean precipitation trend is decreasing. For regions where the precipitation trend is increasing (Figures 4g and 4h) the number and duration of pluvial events fall almost exclusively within the upper right quadrant (95–97%) regardless of the severity. The same is seen in locations where the precipitation trend is decreasing (Figures 4i and 4j, 89–98%). However, pluvial events are not exclusively occurring in regions with a wetting trend (and vice versa for droughts), with the largest changes in the most severe events. In locations where precipitation is projected to decrease, over 60% of grid points see severe pluvials that are longer and 40% of grid points see more pluvials (Figure 4l). For severe drought events in regions where precipitation is projected to accurate the projected to see an increase in duration and 25% are projected to see an increase in number of events.

Figure 5 shows regions where changes in pluvial and drought characteristics are in the opposite sense to the precipitation trend in the multimodel mean CMIP5 simulations for the severe events. For example, longer pluvials where the precipitation is projected to decrease. Results for moderate events are consistent and shown in Figure S6. For the number (Figure 5a) and duration (Figure 5b), the signal is predominantly in regions where the model agreement on the sign of the trend is low, across the central United States for example. Increased duration of pluvial events is projected in the Amazon, South Africa, and Mediterranean where models agree on decreasing precipitation in the future. The most spatially extensive signal is for the intensity of events (Figure 5c), where 70% (Table S3) of grid points show this opposite relationship. These results suggest that even in regions where models strongly agree on annual average precipitation decreases in the future (e.g., South Africa, the Southwest United States, and portions of Brazil), severe pluvial event are projected to increase in intensity. The opposite is seen in regions where precipitation is projected to decrease (e.g., northern North America, East Africa, and Asia), where the intensity of severe and moderate drought events are projected to increase.

4. Summary and Conclusions

The ability of historical simulations from 24 coupled ocean-atmosphere global climate models from CMIP5 to simulate number, duration, intensity, and severity of pluvial and drought events has been assessed. Comparing two observational data sets of precipitation (CRU and GPCP) with 24 historical CMIP5 models, agreement is seen in mean precipitation trends in the historical period but there is less agreement (between the two observational data sets and the models) in the variability of precipitation. This highlights challenges with understanding observed precipitation variability. In the regions of disagreement, such as northern North America and central Asia, there is also less certainty in the simulation of pluvial and drought characteristics in the historical period for moderate and severe events. In the future, the variability in the 6-month SPI is projected to increase globally, suggesting that the characteristics of pluvial and drought events will change.

Generally, wet regions are projected to have more frequent, longer lasting, and stronger pluvials (and vice versa for drought events), which is captured by the severity index. However, there are large portions of the globe that will see simultaneous increases in pluvial and drought event severity. As in Findell and Delworth (2010) the Americas stand out for changes in drought characteristics and the Northern hemisphere midlatitudes for pluvials, but significant changes are projected across most of the globe. Especially for the most severe events, locations exist with more, longer, and stronger pluvials despite a decreasing precipitation trend (and vice versa for drought events in regions of overall wetting). Highest confidence is in increased intensity of pluvial events in regions of decreasing precipitation and increased intensity of droughts in regions of increasing precipitation in future projections.

By using SPI to define pluvial and drought event characteristics, we are considering only water supply (i.e., precipitation input). The SPI does not use evaporation or evapotranspiration in its calculation, so capturing the effect of increased temperature in the future is severely limited. Utilizing the standardized precipitation evapotranspiration index (SPEI; Beguería et al., 2014) includes the role of increased temperatures on moisture demand and availability, which is of particular importance for drought events (e.g., Ficklin & Novick, 2017; Novick et al., 2016), but is used less frequently for pluvials. Results using 6-month SPEI, show greater changes in drought characteristics than for SPI, and smaller changes in pluvial characteristics, but overall, the results are supported by SPEI (Figures S9–S11).

This has important implications for planning and resilience to events as changes in the mean precipitation do not provide information about changes in pluvial and drought characteristics. For example, crop and animal agricultural management Rowhani et al. (2011), reservoir storage and dam release Sakho et al. (2017), and human health (Shively, 2017), among others, have all shown to be affected by changing precipitation variability. Changes in drought characteristics are consistent with those from prior studies, but with changes in pluvials at least as large and widespread as changes in droughts, it is necessary to continue to increase mechanistic understanding on the wet side of extended precipitation extremes.

References

Anderegg, W. R., Kane, J. M., & Anderegg, L. D. (2013). Consequences of widespread tree mortality triggered by drought and temperature stress. *Nature Climate Change*, 3(1), 30.

Auld, H., Maclver, D., & Klaassen, J. (2004). Heavy rainfall and waterborne disease outbreaks: The Walkerton example. *Journal of Toxicology* and Environmental Health, Part A, 67(20-22), 1879–1887.

Acknowledgments

The author thanks two anonymous reviewers for comments that greatly improved the manuscript. This research is supported by the National Science Foundation grant ICER-1663840. We acknowledge the World Climate Research Programme's Working Group on Coupled Modelling, which is responsible for CMIP, and the climate modeling groups for producing and making model output available. For CMIP the U.S. Department of Energy's Program for Climate Model Diagnosis and Intercomparison provides coordinating support and led development of software infrastructure in partnership with the Global Organization for Earth System Science Portals. GPCC Precipitation data are provided by the NOAA/OAR/ESRL PSD. Boulder, Colorado, USA, from their web site at https://www.esrl.noaa.gov/psd/. CRU data are available from the Climatic Research Unit, University of East Anglia.



Beguería, S., Vicente-Serrano, S. M., Reig, F., & Latorre, B. (2014). Standardized Precipitation Evapotranspiration Index (SPEI) revisited: Parameter fitting, evapotranspiration models, tools, datasets and drought monitoring. *International Journal of Climatology*, 34(10), 3001–3023.

Bronstert, A. (2003). Floods and climate change: Interactions and impacts. *Risk Analysis: An International Journal*, 23(3), 545–557.
Chou, C., Neelin, J. D., Chen, C.-A., & Tu, J.-Y. (2009). Evaluating the "rich-get-richer" mechanism in tropical precipitation change under global warming. *Journal of Climate*, 22(8), 1982–2005.

Christian, J., Christian, K., & Basara, J. B. (2015). Drought and pluvial dipole events within the great plains of the United States. Journal of Applied Meteorology and Climatology, 54(9), 1886–1898.

Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., Friedlingstein, P., et al. (2013). Long-Term climate change: Projections, commitments and irreversibility (pp. 1029–1136). Cambridge, UK and New York: Cambridge University Press.

Cook, B. I., Smerdon, J. E., Seager, R., & Coats, S. (2014). Global warming and 21st century drying. Climate Dynamics, 43(9-10), 2607–2627. Dai, A., & Zhao, T. (2017). Uncertainties in historical changes and future projections of drought. Part I: Estimates of historical drought changes. Climatic Change, 144(3), 519–533.

Duffy, P. B., Brando, P., Asner, G. P., & Field, C. B. (2015). Projections of future meteorological drought and wet periods in the Amazon. Proceedings of the National Academy of Sciences, 112(43), 13,172–13,177.

Epule, E. T., Peng, C., Lepage, L., & Chen, Z. (2014). The causes, effects and challenges of Sahelian droughts: A critical review. *Regional Environmental Change*, 14(1), 145–156.

Ficklin, D. L., & Novick, K. A. (2017). Historic and projected changes in vapor pressure deficit suggest a continental-scale drying of the United States atmosphere. Journal of Geophysical Research: Atmospheres, 122, 2061–2079. https://doi.org/10.1002/2016jd025855

Findell, K. L., & Delworth, T. L. (2010). Impact of common sea surface temperature anomalies on global drought and pluvial frequency. Journal of Climate, 23(3), 485–503.

Govender, N., Trollope, W. S., & Van Wilgen, B. W. (2006). The effect of fire season, fire frequency, rainfall and management on fire intensity in Savanna vegetation in South Africa. *Journal of Applied Ecology*, 43(4), 748–758.

Harris, I., Jones, P., Osborn, T., & Lister, D. (2014). Updated high-resolution grids of monthly climatic observations—The CRU TS3. 10 dataset. International Journal of Climatology, 34(3), 623–642.

Hunt, B. (2011). Global characteristics of pluvial and dry multi-year episodes, with emphasis on megadroughts. International Journal of Climatology, 31(10), 1425–1439.

Kangas, R. S., & Brown, T. J. (2007). Characteristics of us drought and pluvials from a high-resolution spatial dataset. International Journal of Climatology, 27(10), 1303–1325.

Li, Y., Ye, W., Wang, M., & Yan, X. (2009). Climate change and drought: A risk assessment of crop-yield impacts. *Climate Research*, 39(1), 31–46.
Martin, E. R., Thorncroft, C., & Booth, B. B. B. (2014). The multidecadal atlantic SST—Sahel rainfall teleconnection in CMIP5 simulations. *Journal of Climate*, 27, 784–806. https://doi.org/10.1175/JCLI–D–13–00.242.1

McKee, T. B., Doesken, N. J., & Kleist, J. (1993). The relationship of drought frequency and duration to time scales. In Proceedings of the 8th Conference on Applied Climatology (Vol. 17, pp. 179–183). Boston, MA: American Meteorological Society.

NOAA (2016). National centers for environmental information. State of the climate: National climate report for annual 2015. Retrieved from https://www.ncdc.noaa.gov/sotc/national/201513

Novick, K. A., Ficklin, D. L., Stoy, P. C., Williams, C. A., Bohrer, G., Oishi, A. C., et al. (2016). The increasing importance of atmospheric demand for ecosystem water and carbon fluxes. *Nature Climate Change*, 6(11), 1023.

O'Gorman, P. A., & Schneider, T. (2009). The physical basis for increases in precipitation extremes in simulations of 21st-century climate change. *Proceedings of the National Academy of Sciences*, 106(35), 14,773–14,777.

Orlowsky, B., & Seneviratne, S. I. (2013). Elusive drought: Uncertainty in observed trends and short-and long-term CMIP5 projections. Hydrology and Earth System Sciences, 17(5), 1765–1781.

Pendergrass, A. G., Knutti, R., Lehner, F., Deser, C., & Sanderson, B. M. (2017). Precipitation variability increases in a warmer climate. Scientific Reports, 7(1), 17966.

Peters, E. J. (2014). Measuring the severity of dry seasons in the grenadines. West Indian Journal of Engineering, 36(2), 42-50.

Räisänen, J. (2002). CO₂-induced changes in interannual temperature and precipitation variability in 19 CMIP2 experiments. *Journal of Climate*, *15*(17), 2395–2411.

Rind, D., Goldberg, R., & Ruedy, R. (1989). Change in climate variability in the 21st century. Climatic Change, 14(1), 5–37.

Rowhani, P., Lobell, D. B., Linderman, M., & Ramankutty, N. (2011). Climate variability and crop production in Tanzania. Agricultural and Forest Meteorology, 151(4), 449–460.

Sakho, I., Dupont, J.-P., Cisse, M. T., El Janyani, S., & Loum, S. (2017). Hydrological responses to rainfall variability and dam construction: a case study of the upper Senegal river basin. *Environmental Earth Sciences*, 76(6), 253.

Schneider, U., Becker, A., Finger, P., Meyer-Christoffer, A., Rudolf, B., & Ziese, M. (2011). Gpcc full data reanalysis version 6.0 at 0.5: Monthly land-surface precipitation from rain-gauges built on GTS-based and historic data. GPCC Data Rep., doi, 10.

Shively, G. E. (2017). Infrastructure mitigates the sensitivity of child growth to local agriculture and rainfall in Nepal and Uganda. *Proceedings* of the National Academy of Sciences, 114(5), 903–908.

Sun, Q., Miao, C., Duan, Q., Ashouri, H., Sorooshian, S., & Hsu, K.-L. (2018). A review of global precipitation data sets: Data sources, estimation, and intercomparisons. *Reviews of Geophysics*, 56(1), 79–107.

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. *Bulletin of the American Meteorological* Society, 93, 485–498.

Touma, D., Ashfaq, M., Nayak, M. A., Kao, S.-C., & Diffenbaugh, N. S. (2015). A multi-model and multi-index evaluation of drought characteristics in the 21st century. *Journal of Hydrology*, 526, 196–207.

Trenberth, K. E. (2008). The impact of climate change and variability on heavy precipitation, floods, and droughts. In M. G. Anderson (Ed.), Encyclopedia of Hydrological Sciences. John Wiley & Sons. https://doi.org/10.1002/0470848944.hsa211

Wilhite, D. A., & Glantz, M. H. (1985). Understanding: The drought phenomenon: The role of definitions. Water International, 10(3), 111–120.